Author’s response to reviews

Title: Case report: term birth after fertility-sparing treatments for stage IB1 small cell neuroendocrine carcinoma of the cervix

Authors:

Pei-Ying Wu (anna1002ster@gmail.com)
Ya-Min Cheng (chengym@mail.ncku.edu.tw)
Geok Huey New (newgeokhuey@hotmail.com)
Cheng-Yang Chou (chougyn@mail.ncku.edu.tw)
Chun-Ting Chiang (altanasia.chiang@gmail.com)
Hung-Wen Tsai (hungwen@mail.ncku.edu.tw)
Yu-Fang Huang (yufangh@mail.ncku.edu.tw)

Version: 2 Date: 01 Jul 2017

Author’s response to reviews:

Reply to reviewer’s comments

Reviewer 1 (Ahmed A Nassr)

Comments:

Ahmed A Nassr (Reviewer 1): interesting case report The authors addressed the comments appropriately

Reply: Thank you for your comment.

Reviewer 4 (Jesus Salvador Jiménez López, Ph.D, M.D.)

Comments:
Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format.

Please overwrite this text when adding your comments to the authors.

Reply: Thank you for your comments.

Reviewer 5 (Mehmet Baki Şentürk)

Comments:

Thank you for your corrections.

Reply: Thank you for your comment.